Author's response to reviews

Title: Patterns of sexually transmitted infections in adolescents and youths in Dar es Salaam, Tanzania

Authors:

Guerino Chalamilla (gchalamilla@muchs-harvard.org)
Judica Mbwana (jmbwana@muchs.ac.tz)
Fred S Mhalu (fmhalu@muchs.ac.tz)
Eunice Mmari (emmari@muchs.ac.tz)
Mtebe Majigo (mmajigo@muchs.ac.tz)
Andrew B Swai (aswai@muchs.ac.tz)
Willy K Urassa (wurassa@muchs.ac.tz)
Eric Sandstrom (eric.sandstrom@karolinska.se)

Version: 9 Date: 28 December 2005

Author's response to reviews: see over
Dear
Thanks for accepting to publish our paper in your Journal. We appreciate your guidance in finalizing this manuscript

Following your comments we have done the following in the manuscript

1. We have ensured that the Title and abstract are identical in the manuscript and in the submission system
2. We have ensured that the list of the authors in the manuscript are written exactly as they are in the submission system
3. We have removed the key words from the manuscript.
4. We have prepared the four tables using the Table tool in word. They are all divided into cells